

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Espín-Sánchez, José-Antonio; Gil-Guirado, Salvador; Vickers, Chris

Working Paper

La "Doña" è mobile: The role of women in social mobility in a premodern economy

Economic Growth Center Discussion Paper, No. 1081

Provided in Cooperation with: Yale University, Economic Growth Center (EGC)

Suggested Citation: Espín-Sánchez, José-Antonio; Gil-Guirado, Salvador; Vickers, Chris (2020) : La "Doña" è mobile: The role of women in social mobility in a premodern economy, Economic Growth Center Discussion Paper, No. 1081, Yale University, Economic Growth Center, New Haven, CT

This Version is available at: https://hdl.handle.net/10419/243239

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Yale University

EliScholar – A Digital Platform for Scholarly Publishing at Yale

Discussion Papers

Economic Growth Center

7-2020

La "Doña" è Mobile: the Role of Women in Social Mobility in a Premodern Economy

José-Antonio Espín-Sánchez Yale University, jose-antonio.espin-sanchez@yale.edu

Salvador Gil-Guirado Universidad de Murcia, salvador.gil1@um.es

Chris Vickers Auburn University, czvickers@auburn.edu

Follow this and additional works at: https://elischolar.library.yale.edu/egcenter-discussion-paper-series

Part of the Growth and Development Commons

Recommended Citation

Espín-Sánchez, José-Antonio; Gil-Guirado, Salvador; and Vickers, Chris, "La "Doña" è Mobile: the Role of Women in Social Mobility in a Pre-modern Economy" (2020). Yale Economic Growth Center Discussion Paper 1081. https://elischolar.library.yale.edu/egcenter-discussion-paper-series/1081

This Discussion Paper is brought to you for free and open access by the Economic Growth Center at EliScholar – A Digital Platform for Scholarly Publishing at Yale. It has been accepted for inclusion in Discussion Papers by an authorized administrator of EliScholar – A Digital Platform for Scholarly Publishing at Yale. For more information, please contact elischolar@yale.edu.

La "Doña" è Mobile: the Role of Women in Social Mobility in a Pre-modern Economy*

José-Antonio Espín-Sánchez Yale University Salvador Gil-Guirado Universidad de Murcia Chris Vickers Auburn University

Abstract

We use data from marriage records in Murcia, Spain, in the 18th century to study the role of women in social mobility in the pre-modern era. Our measure of socioeconomic standing is identification as a don or doña, an honorific denoting high, though not neccesarily, noble status. We show that this measure, which is acquired over the lifecycle, shows gendered transmission patters. In particular, same-sex transmission is stronger than opposite-sex, for both sons and daughters. The relative transmission from fathers versus mothers varies over the lifecycle, and grandparents may have an effect on the status of their grandchildren.

JEL Codes: D31; J62; N33

Keywords: Social mobility; Spain; Assortative mating; Marriage; Status

^{*}Discussions with Armando Alberola Romá, Metin Cosgel, Neil Cummins, Joseph Ferrie, Tim Guinnane, Naomi Lamoreaux, Jason Long, Enrique Llopis Agelán, José Miguel Martínez Carrión, Laura Salisbury, Angel Pascual Martínez Soto and Cayetano Mas Galvañ have greatly benefited this work. We would like to thank William Giraldo, Kevin Lai, Sarah Merchant, and especially Miguel Ángel Lafuente Navarro for their excellent help. Also we acknowledge financial support from the Economic Growth Center at Yale University. Emails: José-Antonio Espín-Sánchez: jose-antonio.espin-sanchez@yale.edu, Salvador Gil-Guirado: salvador.gil1@um.es and Chris Vickers: czvickers@auburn.edu.

1 Introduction

The study of historical patterns in intergenerational mobility has experienced a resurgence in recent years. The increasing availability of large datasets linking individuals and families over time, coupled with awareness of declines in social mobility in the United States over the past several decades, has led researchers to investigate the degree to which societies vary in intergenerational mobility, as well as why these differences arise (Chetty et al., 2014). The bulk of studies on social mobility focus on the transmission of socioeconomic status (SES) between fathers and sons, with status usually measured by occupation. However, this narrow focus could hide interesting dynamics on the role of women and the evolution of this over time. The correlation in SES between fathers and sons is a function of both the effect of fathers and of mothers on sons. Moreover, most of the work is done for post-industrial societies, with the result that we have an incomplete picture of social mobility in pre-industrial societies, and particularly the importance of the role of women in them.

We use marriage records for the city of Murcia (Spain) during the 18th century to study social mobility and assortative mating. The records contain the socioeconomic status of the bride, the groom, and their parents. We investigate the role that fathers and mothers, independently, have in determining the SES of their sons and daughters. Very little is known about inequality and social mobility in pre-industrial societies, especially for late-industrializing countries (Espín-Sánchez et al., 2019). While historians of the family have examined the topic, they have approached it in a less quantitative way, focusing instead on particular case studies and narrative evidence.¹ By the end of the 18th century, Murcia was experiencing important structural economic changes due to the early stages of industrialization (María Teresa Pérez Picazo, 1979). Results from this period could thus shed light on the differences between early industrializing countries, such as the US and the UK, and

¹Chacón Jiménez and Recaño Valverde (2002) use a cross-sectional dataset (the Godoy's census of 1797) to study the neighboring town of Lorca.

late industrializing countries like Spain.

To measure status, we use whether or not an individual is recorded as a *don* or a *doña*. These are Spanish honorifics denoting individuals who, while not necessarily noble, are of relatively high status. Alfani and Gourdon (2016) use a similar measure, *monsieur* and *signore*, to ascribe high status to godparents for France and Italy respectively. We discuss this measure in detail below, but we note here several salient features. First, being recorded as a *doña* measures the status of women directly, as it is not imputed based on the status of husbands or parents. While such imputations are common, they can obscure gender-specific differences in intergenerational mobility. Second, the status can be gained, but not lost, for a particular individual. Once a person is recorded as a *don* or *doña*, this will be done so consistently in later records. In this sense, the measure is similar to studies of intergenerational correlations in years of education. Third, the honorific is not strictly hereditary, in that some individuals who have parents who are *dons* and *doñas* will have children who are not, and vice versa. The failure of a son to acquire the *don* status of his father does not appear to be based on birth order, as we find no effect of marriage order within a family on intergenerational mobility, as we discuss later.²

This paper contributes to several strands of the intergenerational mobility literature. First, it uses a measure of social status which, to our knowledge, has not been used to measure social mobility before but is available across a variety of places and times and over long periods of time. Second, it studies transmission of status from (and to) women, but using a direct measure of female status, rather than one imputed from the status of women's husbands. Given relatively low levels of information about female occupation or income in historical records, such information is extremely rare. Third, it generates estimates of social mobility (for men and women) for a time and place about which little is known about social mobility, as it predates modern censuses.

The literature in economic history using linking records has grown immensely since

²This might not be surprising given the system of partible inheritance present in Castile (Barrera-González, 1998).

the seminal work of Ferrie (1996). Some examples of this are Long and Ferrie (2007) and Long and Ferrie (2013) for the United States and the United Kingdom before 1950, Parman (2011) for early 20th century Iowa, and Long (2013) for Victorian England. The best quality studies of recent social mobility have been done with administrative Scandinavian data (Jäntti and Jenkins, 2015, Björklund et al., 2017, Landersö and Heckman, 2017) which, unlike the other studies mentioned before, contain the full population and their history, as well as family relations. Moreover, they rely on tax data, which contains several years of income data, so the measurement error in lifetime income is negligible, compared to studies that only use one year of income. However, these studies, by the very nature of the data, cannot go back in the past and rarely use data from before the 20th century. There is also a literature that studies social mobility in European countries using probate records, usually for the 19th century: Harbury and Hitchins (1979) for Britain; Arrondel and Grange (2006) for France; Dribe and Svensson (2008) using local population registers for Sweden; and Santiago-Caballero (2018) using marriage records in Valencia (Spain).

The voluminous literature on intergenerational mobility has generally relied on fatherson links, largely for reasons of data availability. There are two reasons data availability is much greater for males. First, in societies where women lose their last name upon marriage, it is hard to track the matrilineal lineage. Recent studies, however, are using marriage records to find the bride's maiden name (Craig et al., 2019, Dribe et al., 2019), though these generally rely on status of women as measured through that of male relatives. Second, even after linking the mother's family side, individual information on women's socioeconomic status (SES) might not be available. Some authors have opted for computing the correlation between the son and his father-in-law (Santiago-Caballero, 2018, Joana-Maria Pujadas-Mora, 2018). Another solution, particularly associated with Olivetti and Paserman (2015), is to construct pseudo-links, based on first names, between fathers and daughters, in addition to sons.³ In this article, however, we use a direct measure of SES for women,

³There are exceptions to the usual focus only on males, using contemporary data, such as Altonji and Dunn (2000) who study transmission from mothers and to daughters, and Chadwick and Solon (2002), who

which allows us to identify individually the effects of each parent for sons and daughters, and also the degree of assortative mating. In addition to studying the effect of mothers, a number of recent studies have challenged the implicit view that only the status of the parents matters for predicting the status of sons. Long and Ferrie (2018) find that parental grandfathers have predictive status independent of that of fathers, and Olivetti et al. (2018), using pseudo-links, find maternal grandmothers also have predictive power. Our data allows us to link all four grandparents for grooms and all four grandparents for brides.

We begin by examining the transmission of status at the age of children's marriage. Marriage records contain the *don* and *doña* status for the bride and groom, as well as all four parents, at the time of the marriage. This avoids issues arising from of overrepresentation of rare names and incorrect links, both of which can bias estimates. We regress the standardized status of children on that of both their fathers as well as their mothers and find highly persistent status for both men and women. There are substantial differences across gender in how status is transmitted. For grooms, the predictive power of fathers is about twice that of mothers: When regressing the status of sons on both parents, the coefficient on fathers is 0.59 and that on mothers 0.28. For brides, these respective figures are 0.32 on fathers and 0.57 on mothers.

We then link marriage records over time, *i.e.*, the marriage record of a couple and the marriage of one of their children. Notice that, given the Spanish naming convention, we are linking four different words when linking two marriage records—groom's given name and surname, and bride's given and surname—which improves matching accuracy. This linked sample allows us to do two things, both of which are novel. First, they allow us to study the effects of all four grandparents directly, in addition to that of the parents, in both sons and daughters. The coefficients on fathers and mothers for grooms are not much changed when grandparents are added. We show, consistent with Long and Ferrie (2018), that parental grandfathers have predictive status independent of that of parents.

estimate intergenerational elasticities for sons and daughters. More recently, economists have extended the analysis to more complex family dynamics.

The second key advantage of linking marriage records over time is that it allows us to calculate status transmission at two points in the lifecycle. Our estimate without linking uses the status of parents and children at the same date, but at different points in their lifecycle: children at marriage, and parents at their children's marriage. With the linked sample we also study the transmission of status taken at different dates, but at the same point in their lifecycle: young adulthood and middle age. This means we have three measures of status transmission: i) same date; ii) both at young adulthood; and iii) both at middle age. In other words, when we link two samples, we have a measure of status in each sample for the couple linked, one measure when they married and another measure when their child married. The coefficients are lower than in the specification above, but still gendered, particularly when measured young.

The rest of the paper is organized as follows. Section 2 provides historical background on the honorifics *don* and *doña*. Section 3 explains in detail the data sources and how the dataset was created from primary sources. Section 4 presents the main results of the paper. It shows the effects of mothers and father on sons and daughters using the unlinked sample. Using the linked sample it also shows the transmission measured at different points in time, the effects of grandparents, and of household composition. Finally, Section 4 also discusses assortative mating. Section 5 concludes with ideas for future research.

2 Dons and Doñas

The social treatment of *don* and *doña* was originally meant to distinguish the aristocracy (*hidalgos*, literally "son of something") from commoners. Its origins come from the Late Middle Ages, where the king would recognize the value of an individual due to personal merit. In Spain, these merits would typically be linked to some assistance during the *Reconquista*. Over time, however, *hidalgos* were associated with social class, not necessarily with the aristocracy (Pita Pico, 2013). In Spain, the public display of being a *don* has been linked

to the creation of the national identity. Miguel de Cervantes makes his delirious character Alonso Quijano call himself *don Quijote*, thus elevating himself with the elite, present and past. One could become a *don* for different causes: lineage, place of origin, wealth (including purchase of nobility titles), wisdom (such as becoming a doctor), or religious (all priests are automatically *dons*). Therefore, by the 18th century this relation is not that strong, so that some elite people are considered *dons* despite not holding an aristocratic title (Pita Pico, 2013). Appendix A.1 presents more information about *hidalgos* and their prevalence in Spain as a whole. Appendix A.2 and the included tables show the prevalence of *dons* and *doñas* in our data.

Don is a measure of high status in Spain and the Spanish Empire, similar to a *Signore* in Italy, a *Gentleman* in England and a *Monsieur* in France.⁴ There is not a clear consensus in the literature on how this somewhat informal title is transmitted. The literature that studies the meaning of the title is scarce and, to the best of our knowledge, there are no works studying its transmission from the parents to their children. Pérez León (2012) argues that the title transmission in Castile was regulated by the royal decree on July 3, 1611, where the king Philip III established that it should be reserved for bishops, counts, the sons of the high nobility (including illegitimate sons), and the wives and daughters of the low nobility, but not their sons. However, these rules changed over time. The financing needs of the Crown meant that a royal decree on July 3, 1664, by Philip IV, established that the title could be used only after paying a fee: 200 reales for one life (the buyer); 400 reales for two lives (the buyer and his first-born son); and 600 reales in perpetuity (the buyer and his patrilineal descendants). A new royal decree on February 10, 1795, raised the fees to have the title in perpetuity to 1000 reales, raised again to 1,400 reales on 1801 (Bustos Argañarás, 2015). Nonetheless, Soria Mesa (2004) argues that in practice, the title was always up for sale, including bribes to witnesses to certify a noble family origin. Thus the title is less rigid than the official laws would suggest, but it is a good measure of social (elite) status.

⁴The prefixes Von and Van in German and Dutch names respectively, also denoted high or noble status originally.

Moreover, Soria Mesa (2004) argues that, unlike the common opinion, the 17th and 18th centuries in Spain were characterized by high social mobility, where wealth, not birth, played a central role in social status.

One particular case in our data deserves special attention, that of women who appear as *doñas* when they married, even when their parents do not. There are famous cases of women in pre-modern Castile acquiring high status coming from humble origins. Beatriz Galindo, born in 1465, was the first female to get a university degree worldwide. Luisa de Medrano became the first University Professor in the world. Moreover, she replaced the great Antonio de Nebrija as Grammar Professor in the University of Salamanca in 1508. These are, however, rare exceptions. The most common way for women to climb the social ladder would be to marry a high status husband, or to work after marriage in the family business and earn the respect of her community. Beatriz Galindo, mentioned above, was the royal tutor for the daughters of Queen Isabel of Castile. Being a personal tutor for the children of the nobility was another way that a young women would climb the social ladder. Unfortunately, we do not have much information in this regard. Irigoyen López (2012) argues that one key figure for upward mobility was the clergy. Though did not have (legitimate) children of their own, they would play a key role on helping their relatives, such as siblings or nephews, to climb the social ladder. Molina-Puche (2005) emphasizes that this role was particularly relevant for their sisters and nieces. Therefore, some of the cases of upward mobile women that seems paradoxical could be explained by the help they got from a relative in the clergy. Another way, of course, was to become a nun, and ascend in the organization of the Catholic Church. Nuns, however, would not appear in marriage records.

We have 84 cases (out of 18,175) of fully upwardly-mobile women in our dataset, *i.e.*, brides that appears as *doña* when both of their parents and the groom are not *don*. We looked intensively for them in the primary and secondary sources. In fourteen of the cases, the parents come from outside the city. Therefore it is plausible that they were of high

status, but the local priest would not know them well enough as to classify them of high status, even if he did not hesitate to classify the bride as *doña*. In eight of the cases, we could not find any information on the parents, but we have found information on the bride. It is uncertain whether the parents were of high status in this case. In 56 of the cases, we could not find any information, in primary of secondary sources, regarding the bride or her parents. For the remaining six cases, we could find detailed information both on the parents and the bride. Two of these cases exemplify this issue in general. The first case is that of *Doña* Ginesa Martínez, who married José Antonio Oliver on December 25, 1698. She was an important landowner in the neighboring town of Moratalla, due to a previous marriage in 1685 with *Don* Miguel de Medina (Lisón Hernández, 1989). The second case is that of *Doña* Tomasa Galiana, who married Antonio Galvache on February 27, 1755. His father, Onofre Galiana, appears as owner of a flour mill in the city of Murcia on 1753 (Cremades Griñán, 1981). Therefore, it seems that he was wealthy enough so that his daughter would be a *doña* at marriage but, maybe due to his occupation as a miller, not high status enough to be himself call *don*.

3 Data

We use data from the city of Murcia, located in southeastern Spain (see Figure 1). Murcia is the historical capital of the Kingdom of Murcia and the current Region of Murcia. Since it is the bureaucratic capital, it has historically drained resources from the rest of the region. Fortunately, Murcia has not suffered much from siege or pillage since the 14th century, which means it has excellent historical sources (Espín-Sánchez et al., 2019). The data come from marriage records in Murcia for the 18th century. These church records exist for the years 1565 through 1910, and some of them are available at FamilySearch (2016). However, not all have been transcribed. Although much of the basic demographic data for the transcribed records is available on ancestry.com, the measure of social status we employ is not transcribed. The key point for our purposes is that the records contain six related individuals, all recorded with a social status as either a *don* or *doña* or not. That is, we observe the names and status of the grooms and brides as well as both of their parents.

The parish records are books created by the local priest of every church/parish. They keep the basic information from the main sacraments performed by the priests in a given parish. The main relevant records refer to death, baptisms, and marriage records. They are useful for genealogical studies and to study social mobility. They are very similar across time and across the Catholic world, especially after the Council of Trent (1563) imposed homogeneity in the sources and made it compulsory for priests to keep accurate records (Chacón Jiménez, 1987, Pérez Ortíz et al., 2017). While the sources are homogeneous across time and space, their survival is heterogeneous. Even within Spain, the sources have not survived in many cities, as a consequence of thefts, wars, fires, and other hazards (Pérez Ortíz et al., 2017). Nonetheless, the region of Murcia, and in particular the city of Murcia, has one of the best preserved demographic records in Spain, and in the world. Most of the Murcia parishes preserved most of their data, at least since 1563 (Irigoyen López, 2012, Chacón Jiménez and Chacón Martínez, 2015).

The city of Murcia, like many medieval cities, was restricted in size by its medieval walls. In addition to the walls, the old city was bounded in the south by the Segura River. Figure 1 shows the "old" parishes: Santa María, San Pedro, Santa Catalina, San Antolín, San Juan, San Lorenzo, San Miguel, San Nicolas and Santa Eulalia.⁵ Santa María is the name of the cathedral parish, San Bartolomé is the closest church, and the two parishes merged into a single parish for administrative purposes. We refer to the merged parish as Santa María for simplicity. In Table 1, we can see that Santa María holds a much larger number of records than the other parishes.⁶ The Santa María parish provided sacraments

⁵To the best of our knowledge, this is the first map representing the parishes in the city of Murcia. We constructed the map using a booklet (*Boletín Eclesiástico*) from the Cargatena Bishopric from 1908, pages 89-94 (Obispado de Cartagena, 1909). In the memo, there is a detailed description of the limits of each parish, listing the street names that mark the boundaries. Because some streets have changed names since 1908, we used historical maps of Murcia from 1896 and 1882 (Archivo General Región de Murcia, 2019).

⁶The "new" parishes were created in the early 19th century. They are all attached to the city walls, from

for citizens in the irrigated orchards around the city, before these areas had their own parishes (Henarejos López, 2011). The "old" parishes are clustered around the cathedral and inside the medieval city walls.⁷

Marriage records contain the date and place of the marriage, and the full name of the groom, the bride, their parents, and the witnesses. By full name, we mean that the record includes the honorary *don* or *doña*, which we use to denote elite status. The marriage records also contain the following information: name of the priest; name of the notary; place of residence of the groom, the bride and the witnesses (best men and maids of honor); place of origin of the groom and bride and their parents; whether the groom or the bride are widows, and the name of their deceased spouse; and whether the groom and the bride are relatives, up to third cousins. In Figure 2, we can see how a single record contains the links for both the groom and the bride with their parents, and the social status of each individual. For this reason, we do not need to link records to perform the main analysis. Moreover, the marriage records contain the same information for men and women. In this particular example, we can see how assortative mating is not perfect. The groom and bride have different status. The same is true for the parents of the groom, but the parents of the bride are both from high status. Notice that, whereas the parents of the bride did transmit their high status to their daughter, the mother of the groom could not transmit her high status to her son.

In order to study the effects of grandparents and family composition, as well as the transmission of status at different ages, we link marriage records backwards one generation, as described in the Appendix A.3. We perform an iterative matching procedure, first

the outside and grew concentrically outwards. In this paper, we focus on the old parishes of the city of Murcia. The sources survived for all parishes, except for San Nicolas during 1700-1780. We drop this parish from our main specification. The church of Santa Eulalia was built in 1766. The marriage records for this parish are only available after 1790, so we do not use it in our main specification. The new parishes were created in the 19^{th} century and so are not relevant to our analysis.

⁷The medieval wall was built in the 12th century, during the Islamic occupation (Garcia Antón, 1993). After 1492, with the union of Castile and Aragon, and the conquest of Granada, the region no longer had a land border with a hostile enemy. This *Pax Iberica* meant that a defensive structure like the city walls was no longer necessary (Jiménez Castillo and Sánchez González, 2004). Nonetheless, parts of the wall were preserved as a protection from floods until the 19th century (García-Tornel, 1997).

matching exactly, then after performing some standardization of names, and then after a phonetic regularization. In a few cases, we have multiple matches, but in the vast majority of these cases, the relevant variables, that is the statuses of the marrying couple and their parents, were identical between the multiple matches. In this case, we included the information from one of the matches.

Note that in addition to expanding the set of relatives, linking gives us two measures for parents and children, at different ages. Consider for concreteness a son, whom we label in generation as G3, a father (G2), and a grandfather (G1). We link the son's marriage record at time T2 to that of his father at earlier time T1. Note this gives us a measure of transmission from fathers to sons recorded when both married, that is, in young adulthood. This measure uses the groom (G3) at T2 and the groom (G2) at T1. Additionally, there is a measure of transmission of status recorded when one of their children married, which we denote as at middle age. This is the father (G2) at T2 and the father (G1) at T1.

4 Results

We now describe the transmission of social status as measured by being recorded as a *don* or *doña* in the marriage records. Several important stylized facts emerge. As mentioned earlier, once an individual becomes a *don* or *doña*, they maintain such status all their lives. In that sense, this measure of social status is similar to years of education. Our first set of results refers to unlinked data. We find that the transmission of social status by this measure is relatively persistent and that the transmission differs by gender, with fathers having a stronger effect on sons and mothers on daughters.

We can link together two marriage records for each couple: when they married and when the parents of either the bride or the groom married. This means that for the linked sample, we have two measures of social status at different points in their lifetime, and also that we can measure the effect of grandparents on predicting social status. We find that gendered effects on the transmission of status are lower for grooms, but not for brides, when we measure status at older ages by using linked marriage records. Finally, we find that grandparents' status has a direct effect on that of their grandchildren, contrary to models of social transmission status which are first order Markov.

4.1 Intergenerational Transmission of Status

For all of the following specifications, we standardize the measure of status, a binary indicator for status as a *don* or *doña*, to have constant mean and variance. We begin the analysis in Table 2.A, first examining the transmission of social status to grooms. All regressions throughout contain fixed effects for the parish in which the child was married and standard errors are clustered at the parish level. The first column uses only the status of the father to predict the status of the son, in line with the most basic test of social mobility. We find an elasticity of 0.79, high but in line with the historical record as suggested by Clark (2014). The equation used by Clark (2014) and most of the literature, corresponding to the first column, is the following

$$S_i = \tilde{\beta}_F^S F_i + \tilde{\varepsilon}_i^S \tag{1}$$

where S_i is the status of the son and F_i is the status of the father. Unlike most of the literature we also have information on the status of the mother, denoted as M_i . The coefficient in the second column, using the mother only as the predictor, is slightly lower, at 0.71. Moreover, we can decompose the effect of both the father and the mother using the following equations, where D_i is the status of the daughter:

$$S_i = \beta_F^S F_i + \beta_M^S M_i + \varepsilon_i^S \tag{2}$$

$$D_i = \beta_F^D F_i + \beta_M^D M_i + \varepsilon_i^D \tag{3}$$

In the third column, we predict the groom's status using the status of both of his parents to disentangle the relative contributions of each to the social status of the child. Both the mother's and the father's status independently predict the social status of the sons, with the effect of the mother about half the magnitude of that of the father.

At this point, it is useful to consider the implied correlation in status of these models. Note that if the true model of status is the one in equation 2, but we instead estimate equation 1, then $\tilde{\beta}_F^S = \beta_F^S + \rho \beta_M^S$, where ρ is the correlation in status between fathers and mothers. Therefore the correlation between the father and son status, $\tilde{\beta}_F^S = 0.790$, can be decomposed into the direct effect of the father $\beta_F^S = 0.586$ and the indirect effect that is equal to the degree of assortative mating $\rho = 0.734$ times the direct effect of the mother $\beta_M^S = 0.278$. Here, $\rho = 0.734$ corresponds to the correlation between the parents' status. The same decomposition is possible when looking at the correlation between mothers and sons, or for either parent and daughters.

We then look at the social status of the daughters, based on the status of their parents, in the first through third columns of Table 2.B. Looking only at column (1), the coefficient is 0.72, relatively similar to that for fathers and sons, though slightly lower. In column (2), predicting the daughter's social status based on that of her mother, the figure is 0.79, essentially identical to that of fathers and sons. When the father and mother of the bride are considered separately, in the sixth column, the coefficients are 0.32 and 0.56. That is, for brides the social status of the *mother* has roughly twice the effect of the father, the opposite of what held for grooms.

Table 2 shows several important facts for social status transmission. Transmission to daughters appears quite similar to that of sons, in the sense that the coefficient is high. However, the *relative* contributions to the status of children from each parent is different for sons versus daughters. In particular, the same-sex transmission of status is roughly twice as strong as that of the opposite-sex parent. Second, the coefficients when looking at each parent fit a similar pattern of same-sex vs opposite sex transmission. That is, the

father-son coefficient and mother-daughter coefficient are virtually identical. Similarly, the mother-son coefficient and father-daughter coefficients are both very similar. Notice that these measures are not intergenerational elasticities per se. The variables are measured at the same date, but at different points in the life-cycle for the parent and the children. Instead, they show that status *as measured at young adulthood* is strongly dependent on status of parents *as measured at their middle age*. Moreover, the transmission of that status is gendered: Fathers matter more for sons, and mothers for daughters. In the next subsection, we discuss how the measures change if we look at variables for parents and children at the same point in their life-cycle.

In Figure 3, we show the results from estimating equations (1) and (2) for sons and daughters separately by decade. When estimated by itself father-son transmission is steady, *i.e.*, the empirical correlation between father and son status is stable. However, we see some evidence for a change in gendered patterns over time, with the mother becoming relatively more important to the status of the son than the father. This is important and underscores our contribution of using a direct measure of socioeconomic status for the mother. Without this direct measure we would get a biased measure in the effect of the father on the son, as discussed above. Moreover, we would also get a biased measure about the evolution of social mobility over time.

In historical studies, the best method for assigning women's status is by using pseudolinks based on names, and assigning to a woman her father's status (Olivetti and Paserman, 2015). Olivetti and Paserman (2015) acknowledge that their method could create biased estimates, but it is useful to see changes over time, if the bias is constant over time. The results in Figure 3 shows that, in the case of 18th century Murcia, the bias does change over time. This implies that one should also be cautious when drawing conclusions about changes over time based on the assumption of a constant bias. If the role of women is changing over time, then so is the bias.

4.2 Transmission of Status in Middle Age and Young Adulthood

In this subsection, we analyze the transmission of status from parents to children again, but after having matched the marriage records to the marriage record of a previous generation. In other words, given the parents of a groom or a bride in a marriage record, we link this marriage record to the one in which the parents were married. We discuss the details of the matching process in Appendix A.3. Note that the regressions above consider the relationship between the status of an individual at marriage and the status of his or her parents at the child's own marriage. However, we may be more interested instead at the transmission of status as measured at a consistent age, which we discuss below.

The analysis in the previous subsection has the advantage that the status of the parent and the child are measured at the same date and, more importantly, it does not require linking. In addition to avoiding linking, measuring status for parents and child in the same document has other advantages. By seeing what elements of status are transmitted to children at marriage versus when it is measured later in life, we get further information on the life-cycle nature of this transmission. Such a measure does have some disadvantages as well. First, the ages at which status is measured are different for the parent and the child, in that it is the age of marriage for the children and the age of their children's marriage for parents. This is somewhat difficult to compare with more traditional measures of intergenerational status transmission. Second, we may be more interested in status during late adulthood. If the status of children is primarily based on that of parents at marriage, but subsequent young and middle adulthood causes the children to develop status independently, then the above specifications will overstate the degree of persistence in status. Rather than viewing either the measures of the transmission at younger or older ages, or alternatively measuring at different ages, as the "correct" model of status transmission, we view each of them as providing information about the nature of the intergenerational persistence of status.

The linked sample provides us with two potential measures of status measured at con-

sistent ages: status of the parents of the marrying couple for both the older and younger generation, *i.e.*, in middle age; and the status of the marrying couples themselves for both generations, *i.e.*, both at young age. We do this exercise for both brides and grooms, and this gives a separate estimate of status transmission. The results for status of grooms measured at middle age for both generations are in the first three columns of Table 3.A. The coefficient is substantially lower, at 0.62, than in the regression in Table 2.A. Perhaps more interestingly, the coefficients on men and women are much more similar when measured at the same age. Status thus appears more "persistent" when measured at the time of marriage, in that sons most closely resemble their middle age fathers at the time of the son's own marriage. The same exercise is repeated for status measured at age of marriage for both generations (that is, at a young age for both fathers and sons) in the fourth through sixth columns of Table 3.A. Here, the gender differences observed above in the first regressions are more apparent: The relative contribution of fathers to social status is much higher than that of mothers, compared to when status is measured in middle age.

We then perform the same exercise for brides, with the results in Table 3.B. When we look at the transmission of status as measured in middle age, we find somewhat lower coefficients than for grooms and considerably lower than for age-at-marriage transmission. In particular, in column (1) the father-daughter coefficient is measured to be 0.43, whereas the coefficient for mother-daughter is higher at 0.49. For brides in comparison to grooms, the gendered difference is sharper when status is measured at middle age. When we estimate both fathers and mothers together in column (3), the coefficient on the father is 0.22 and that on the mother is 0.37. The more surprising finding is when looking at status measured as young adults. Here, the gendered coefficients are *reversed*: Status of the father is more predictive of status of the daughter than the status of the mother.

One explanation for this surprising result when using the measure of status as young adults is that the measure of status may contain some error. As explained in Section 2, the status is acquired over time. This means that when measuring the status at middle age,

all men and women that would eventually become *don* have already done so. When measuring status at young adults, however, many individuals that would become *don* have not already done so, and are thus measured as low status. Measurement error alone, however, cannot explain all the results. In particular, measurement error would mean that the coefficients for middle age in Table 2 should be larger than those in Table 3. But this is not what we see in the data.

Comparing the results in Tables 3 and 2, we see that the coefficients and the R-squared are always smaller in Table 2. This is consistent with the idea that what is being transmitted is not something intrinsic, but rather something temporary. What matters in determining your status when you married is not so much the status that your parents had when they married, but their status now. Notice that the results in Tables 3 and 2 for status measured at young age have the same variable for the children, and a slightly different variable for the parents. For parents who were always low status, the two measures are the same. For parents who were always high status, *i.e.*, were already high status when they married, the two measures are also the same. The only difference between the two variables is then for parents that were not high status when they married, but were high status by the time their children married. Because the coefficients are larger in Table 3, this implies that the transmission is stronger precisely for those parents which changed status.

A criticism of earlier work on social mobility, associated with Clark (2014), is that occupation or income may fail to capture an underlying "status," or "social competence," with greater persistence than occupational status. The true trait to be transmitted would be status, and occupation would be an imperfect measure of status, leading to attenuation bias when measuring social mobility. In other words, occupation/income is only one component of status. Without information on the other components, the intergenerational transmission equation is misspecified, and in particular attenuation bias leads to estimates of intergenerational mobility showing incorrectly large amounts of social mobility. Clark (2014) used surname-based pseudo links to study social mobility in different countries, often over long time spans. Clark (2014) argues that using surnames better captures a true underlying measure of social competence which is highly persistent, with an intergenerational elasticity of 0.7-0.8 across places and times. We use a direct measure of SES, rather than income or occupation, so we have less or no attenuation bias. Moreover, it is measured at the individual, rather than the group level as with names or surnames in Clark (2014), so the results could be interpreted as individual mobility rather than group mobility (Solon, 2018). Our results suggest that even a direct measure of status may show lower persistence than surname-based measures if information about women's status is directly available. In Appendix A.4, we calculate the estimates for intergenerational mobility using the surname method of Clark (2014).

4.3 Transmission of Status from Grandparents

In this subsection, we examine the effect of the status of grandparents as well as parents by using the matched records mentioned above. First, we examine the transmission of status to sons in Table 4.A. The first column duplicates Table 2.A, but for the linked sample. There is a slightly higher rate of persistence in these individuals than in the records as a whole. Note that in the specifications here, the status of the child (the dependent variable) is measured at young age, but the status of the parents as well as the grandparents are measured at middle age, *i.e.*, the same specification as in Table 2.⁸

In column (2), we regress the status of grooms on that of their fathers and their paternal grandfathers. Contrary to the first order Markov process model of Clark (2014), we find evidence for an independent effect of paternal grandfathers on status of sons. This is consistent with the findings of Long and Ferrie (2018), which looks at the transmission of status from fathers and paternal grandfathers. This result suggest that their model would be correct here, and that the transmission of status is second order (non-Markovian). There remains a grandparent effect here even when including the status of the mother in the re-

⁸An alternative here would be to use the status of the parents when they are young, which is also observed.

gression in column (3), suggesting that the effect of paternal grandparents in Long and Ferrie (2018) is not an artifact of missing information about the mother's status. We then include all grandparents in the regression. To the best of our knowledge, this is the first empirical study in a pre-industrial society that looks at the effects of all four grandparents. The results in column (4) suggest that among all the grandparents, only the paternal grandfather has a statistically significant effect. Notice that the effect is significant even controlling for the parents and the other grandparents. Moreover, the coefficients on the parents change very little from the previous columns. In Table 4.B, we repeat this exercise for brides. Comparing the first column to the corresponding column in Table 2, in the matched sample the coefficients on fathers and mothers are similar to those in the total population. Unlike the results for grooms, there is little evidence of a grandparent effect for brides. This might not be surprising given that the mobility mechanisms for men and women are different in this society. The coefficients for fathers and mothers are virtually unchanged when we add grandparents to the regressions.

To summarize, we find a role for grandparents in explaining the social status of children, as measured by status at marriage. We find this is true even when considering the status of women separately, suggesting that this grandfather effect is not capturing some latent variable of "status" which the status measures for fathers imperfectly captures. However, this is more true for grooms than brides. For women, there is little evidence of transmission of status from grandparents.

4.4 Family Composition

With the linked marriages we can also examine the effect of family composition on the transmission of social status. In particular, we can observe the number of siblings, both own sex and other, that are in the linked sample, as well as the order in which individuals within the family marry. We note several important caveats to our analysis. First, we only observe linked individuals, so, for example, the "first to marry" among our sample may

not actually be first to marry, if for whatever reason the earlier marrying individual was not linked. Second, we do not observe *birth* order, only *marriage* order. While we believe the results are informative about the effect of within-family composition, they are not identical to observing birth and thus the effects of, for example, rules about inheritance. Third, we do not observe individuals who never marry at all, such as sons who enter the priesthood, even setting aside those who fail to link.

We regress the status of grooms on that of their parents as well as various family composition characteristics, with the results in Table 5. In these regressions, we use the status of parents as measured at the time of their child's wedding, similarly to Table 2. Column (1) in Table 5 just replicates the regression in column (1) on Table 2, but for the linked sample. In column (2) of panel A, we see that the number of siblings has no effect on the status of the groom. This remains true when splitting the siblings into own or other sex siblings in columns (3) and (4). In column (5), being the first groom to marry, an imperfect proxy for being the oldest son in the family, also has no effect on the transmission of social status. This remains the case when including the number of siblings in column (6). This might not be surprising given that Castile has had partible inheritance since the times of the Visigoths. The rules were formalized by the Laws of Toro (1505).⁹

In Table 5.B., we perform the same exercise for brides. Again, the coefficients on the father and the mother change only slightly, and the results for the new variables are all small in magnitude for family composition variables across all specifications. To summarize, when looking at transmission at own marriage with parents status measured at the time of the child's marriage, there is no evidence of any effect of family structure on social mobility.

Due to the incomplete linking, one might be worried that our sample is selected, and that selection is driving our results. One could be concerned that the youngest sisters in high status households are less likely to marry, and that would bias the results. In partic-

⁹Inheritance rules in the Kingdom of Aragon, also in Spain, prescribed primogeniture. It would be interesting for future research to replicate our analysis with data from the Kingdom of Aragon.

ular, if daughters of high status parents were likely to enter the Church rather than marry, for example if families saved resources to increase the eligibility of one daughter, then we would observe only "status preserving" (high status) marriages. The same story could be told for sons, with brothers being pushed into the priesthood, masking downward mobility. In Table 6, we show the number of brothers and sisters within the linked sample for both high status and low status parents. Note that more brides relative to grooms are linked, which goes against the idea that sisters were less likely to marry than brothers. Moreover, individuals have more sisters than brothers in the linked sample. For the same reason, there are more brothers than sisters linked, in each category. For individuals with low status parents, there is no difference in the number of either brothers or sisters between grooms and brides. For high status individuals, however, we have fewer siblings within the linked sample in general. Moreover, there is a marked difference by sex in the number of siblings. That is, grooms have fewer siblings, both brothers and sisters, than do brides. However, there does not appear to be much difference in the relative gap between the number of brothers and sisters of high vs low status brides: High status brides have 0.26 more sisters than brothers, and low status 0.23. This suggests that while family size differs between high and low status, it does not do so in a way that deferentially affects "the second sister" of a family. Similarly, high status grooms have 0.09 more sisters than brothers, 0.14 for low status grooms, a relatively small difference, again suggesting a lack of attrition based on individuals leaving marriage.

4.5 Assortative Mating

In addition to the results regarding social mobility, the data allows to study assortative mating. In Table 7.A, we predict the status of the groom using information about the bride and her parents in a regression, in the same way that the social mobility regressions were defined above. The goal of this exercise is to understand the process of assortative mating, *i.e.*, how the status of the groom and his parents is correlated with the status of his bride,

and whether the other individuals in the family have additional predictive power. In column (1), we see a high correlation between the statuses of the groom and bride, with the correlation at about 0.80. The degree of assortative mating is high, but is far from perfect. This is precisely what allowed us to independently identify the effects of the mother and the father earlier. We now proceed to investigate further the marriage market by looking at the status of the groom and bride parents.

For pre-industrial Murcia, Chacón-Jiménez and Molina-Puche (2004) argue that the parents of the groom and bride try to find a match that is suitable to the human and physical capital of their child. In other words, there is an active mechanism for assortative mating. The main factors here are the prospective match family status and prestige as well as the potential inherited wealth. This is consistent with Molina-Puche (2005) who argue that, for other towns in the region, the local elites use marriages with other families of the elites as a way to consolidate their power.¹⁰

When we regress the status of the groom on that of the bride and her father, we see a significant effect of the status of the bride's father into the status of the groom. This might suggest that assortative mating is non-Markovian. In other words, that the status of the previous generation (parents) have predicting power on the status of the bride's groom. Including the bride's mother individually shows a smaller effect, that is only marginally significant. The results in columns (4), which include both of the bride's parents, seem to suggest that, even conditional on the status of the groom, parents play an important role in the "marriage market." It is important to remark here that, similarly to social mobility, we do see a pronounced gendered pattern. The status of the bride's father, in addition to the status of the bride, seems to have a high predictive power regarding the status of the groom. The fact that it is the bride's father status, and not her mother, which correlates with the groom's status seems to suggest that there is a social relation between the two men

¹⁰McCaa (1984) argues that, for the case of Mexico in the late 18th century, social status was the key determinant for both the timing of marriage and the social status of the bride. For brides, however, there was no relation between their status and the status of the husband, and the timing was mostly determined by the grooms age, with larger age differences for higher status grooms.

(groom and bride's father). In other words, as a bride, if your father has high status, you are more likely to marry with a high status groom, even conditional on your own status at marriage. This is consistent with the bride's father using his network to "find" a suitable husband for her daughter.

We now turn our attention to Table 7.B. The regression here is analogous to the one above: We predict the status of the bride based on the status of the groom and his parents. There we can see a similar picture to the one in Table 7.A. Individually, both the father and the mother of the groom have a significant effect on the status of the bride. Similarly to our results earlier, the effect is gendered. Now it seems that is the status of the groom's mother that affects the bride's status. Again, this is consistent with a gendered marriage market where the groom's mother could more easily find a high status wife for his son, if she herself is of high status.

The picture here of assortative mating is complex. It shows that the status of the parents affect the status of the spouse, even conditional on the spouse's status. Moreover, it seems that mothers find wives for their sons and fathers find husbands for their daughters. There are several implications of these results for understanding marriage markets in pre-industrial Murcia. To understand marriage markets we need to look beyond the status correlation among spouses. Moreover, regarding social mobility, our results raise doubts about whether or not lack of information about women in a study of social mobility can be corrected in an obvious way. Even if one were to have information about the degree of assortative mating as measured by the correlation between husbands and wives, it appears in this data that the status of grooms and brides depends additionally on the *parents* of the spouses. Given that the status of parents is correlated with children, this is further evidence that grandparent effects may exist, and moreover may exist through the effect the status of extended families on the parents themselves through marriage, not only through "direct" effects on grandchildren. Understanding better the effects that each relative have on the status of sons and daughters is then an important avenue for future research.

5 Conclusion

In this paper, we use a community-based measure of socioeconomic status that exists in areas with Spanish origins: The use of honorific titles *don* and *doña*. This measure, at its core, simply measures whether or not this individual has high status as understood by the community, in this case Murcia (Spain) in the 18^{th} century. What is virtually unique about this measure is that it exists for both men and women, allowing us to study the transmission of status from women and to women. We found that the role of mothers in transmitting social mobility is important, especially for daughters. Moreover, we see that the relative importance of mothers and fathers in transmitting their status to their children changed during the 18^{th} , so that by the end of the century the effects of mothers and fathers are similar for both sons and daughters. Further research is needed to understand this change.

Although these particular records were not linked to records which contain income or occupation, a natural extension of this research would be to see how well these measures correlate to measures based on occupation or income. Given Spanish naming conventions as well as the relatively large numbers of surviving records from church archives and censuses, as well as increasingly complete family trees from genealogical efforts, it should soon be feasible to compare these measures, and many of these church records will contain information on honorifics. Such comparisons of income or wealth based measured with "direct" measures of social status can be made across a wide range of places and times, as many societies have honorific titles similar to those we use. The widespread availability of records containing honorific titles will open up avenues for further study of social mobility, and in particular the role of women in it, in many areas in Spain and Latin America. More systematic studies of status and its effects are feasible and the study of intersection of race, class, and gender in social mobility based on acquired status seems particularly fruitful as an setting for future work.

References

- ABRAMITZKY, R., L. P. BOUSTAN, AND K. ERIKSSON (2014): "A Nation of immigrants: Assimilation and Economic Outcomes in the Age of Mass Migration," *Journal of Political Economy*, 122, 467–506.
- ALFANI, G. AND V. GOURDON (2016): "Las familias y la elección de padrinos y madrinas de bautizo en la Europa católica en la Edad Moderna. Balance y perspectivas de investigación," *Revista de Historia Moderna. Anales de la Universidad de Alicante*, 34, 23–42.
- ALTONJI, J. G. AND T. A. DUNN (2000): "An intergenerational model of wages, hours, and earnings," *Journal of Human Resources*, 221–258.
- ARCHIVO GENERAL REGIÓN DE MURCIA (2019): "Plano General de Murcia de 1896 en el que se delimita el casco de la población en el perímetro encerrado por medio de una linea roja," in *PLANERO*,185/4/1.
- ARRONDEL, L. AND C. GRANGE (2006): "Transmission and inequality of wealth: an empirical study of wealth mobility from 1800 to 1939 in France," *Journal of Economic Inequality*, 4, 209–232.
- BARRERA-GONZÁLEZ, A. (1998): "Domestic succession, property transmission, and family systems in the agrarian societies of contemporary Spain," *The History of the Family*, 3, 221–246.
- ВJÖRKLUND, A., M. J^{*}ANTTI, AND M. NYBOM (2017): "The Contribution of Early-life Versus Labour Market Factors to Intergenerational Income Persistence: A Comparison of the UK and Sweden," *Economic Journal*, 127, 71–94.
- Bustos Argañarás, P. (2015): "Antroponimia Hispanoamericana. Real Academia Matritense de Heráldica y Genealogía,".
- CHACÓN JIMÉNEZ, F. (1987): Familia y Sociedad en el Mediterraneo Occidental. Siglos XV-XIX, EDITUM.
- CHACÓN JIMÉNEZ, F. AND A. CHACÓN MARTÍNEZ (2015): "Hogar y Familia: corrientes interpretativas y realidades sociales. Los ejemplos de movilidad de la población (1771) y movilidad social (1797) en Lorca," *Revista de Demografía Histórica*, 33, 29–63.
- CHACÓN-JIMÉNEZ, F. AND S. MOLINA-PUCHE (2004): Familia, élite local y movilidad social en el Norte del Reino de Murcia. El ejemplo de Yecla en el siglo XVII, Universidad de Castilla-La Mancha, vol. 1 of Declinación de la Monarquía Hispánica en el siglo XVII, 677–694.
- CHACÓN JIMÉNEZ, F. AND J. RECAÑO VALVERDE (2002): "Marriage, work, and social reproduction in one area of southern Europe at the end of the 18th century: Lorca (1797)," *The History of the Family*, 7, 397–421.
- CHADWICK, L. AND G. SOLON (2002): "Intergenerational income mobility among daughters," *American Economic Review*, 92, 335–344.
- CHETTY, R., N. HENDREN, P. KLINE, AND E. SAEZ (2014): "Where is the land of Opportunity? The Geography of Intergenerational Mobility in the United States," *Quarterly Journal of Economics*, 1553–1623.

- CLARK, G. (2014): *The Son Also Rises: Surnames and the History of Social Mobility*, Princeton University Press.
- CRAIG, J., K. ERIKSSON, AND G. T. NIEMESH (2019): "Marriage and the Intergenerational Mobility of Women: Evidence from Marriage Certificates 1850-1910," *Working Paper*.
- Скемадея GRIÑÁN, C. M. (1981): "Gastos municipales en el concejo de Murcia (1749-1759)," Anales de la Universidad de Murcia, 39.
- DEWALD, J. (1996): The European Nobility, 1400-1800, vol. 9, Cambridge University Press.
- DRIBE, M., B. ERIKSSONA, AND F. SCALONE (2019): "Migration, marriage and social mobility: Women in Sweden 1880-1900," *Explorations in Economic History*, 71, 93–111.
- DRIBE, M. AND P. SVENSSON (2008): "Social Mobility in Nineteenth Century Rural Sweden A Micro Level Analysis," *Scandinavian Economic History Review*, 56, 122–141.
- Espín-Sánchez, J. A., S. GIL-GUIRADO, W. D. GIRALDO-PAEZ, AND C. VICKERS (2019): "Labor income inequality in pre-industrial Mediterranean Spain: The city of Murcia in the 18th century," *Explorations in Economic History*, 73.

FAMILYSEARCH (2016): "España, registros parroquiales y diocesanos, 1307-1985,".

- FERRIE, J. P. (1996): "A new sample of males linked from the public use microdata sample of the 1850 US federal census of population to the 1860 US federal census manuscript schedules," *Historical Methods: A Journal of Quantitative and Interdisciplinary History*, 29, 141–156.
- GARCIA ANTÓN, J. (1993): Las murallas medievales de Murcia, Real Academia Alfonso X el Sabio/Universidad de Murcia.
- GARCÍA-TORNEL, F. C. (1997): "Ciudad y río en la cuenca baja del Segura," Areas: Revista internacional de ciencias sociales, 145–154.
- Goñi, M. (2015): "Assortative Matching and Persistent Inequality: Evidence from the World's most exclusive Marriage Market," *Working paper*.
- HARBURY, C. AND D. HITCHINS (1979): Inheritance and Wealth Inequality in Britain, Allen and Unwin.
- HENAREJOS LÓPEZ, J. F. (2011): "La consanguinidad en la ciudad de Murcia (1700-1780): una aproximación a su estudio," *Norba. Revista de historia*, 24, 53–62.
- INE (2019): "Censo de Floridablanca. Edición Facsimil,".
- IRIGOYEN LÓPEZ, A. (2012): "Ecclesiastical godparenthood in Early Modern Murcia," in *Spiritual Kinship in Europe*, 1500-1900, Palgrave Macmillan, London, 74–95.
- JÄNTTI, M. AND S. P. JENKINS (2015): "Income Mobility," in *Handbook of Income Distribution*, Elsevier BV, 807–935.
- JIMÉNEZ CASTILLO, P. AND M. SÁNCHEZ GONZÁLEZ (2004): "Un tramo de la muralla medieval de Murcia y el área urbana adyacente. El solar de calle Sagasta, esquina con calle brujería," *Memorias de arqueología*, 12, 481–502.

- JOANA-MARIA PUJADAS-MORA, GABRIEL BREA-MARTÍNEZ, J.-P. J. S. A. C. (2018): "The apple never falls far from the tree: siblings and intergenerational transmission among farmers and artisans in the Barcelona area in the sixteenth and seventeenth centuries," *The History of the Family*, 23, 533–567.
- LANDERSÖ, R. AND J. J. HECKMAN (2017): "The scandinavian fantasy: The sources of intergenerational mobility in denmark and the us," *The Scandinavian Journal of Economics*, 119, 178–230.

LISÓN HERNÁNDEZ, L. (1989): "Estampas de Moratalla en 1685," Murgetana, 79.

- LONG, J. (2013): "The surprising social mobility of Victorian Britain," European Review of Economic History, 17, 1–23.
- LONG, J. AND J. FERRIE (2007): "The path to convergence: intergenerational occupational mobility in Britain and the US in three eras," *The Economic Journal*, 117, C61–C71.
 - (2013): "Intergenerational Occupational Mobility in Great Britain and the United States Since 1850," *American Economic Review*, 103, 1109–1137.
 - (2018): "Grandfathers matter (ed): occupational mobility across three generations in the US and Britain, 1850-1911," *The Economic Journal*, 128, F422–F445.
- MADRAMANY Y CALATAYUD, M. (1788): Tratado de la nobleza de la Corona de Aragón: especialmente del reyno de Valencia, comparada con la de Castilla, para ilustración de la Real Cedula del Señor Don Luis I de 14 de agosto de 1724, Josef y Tomas de Orga.
- MARÍA TERESA PÉREZ PICAZO, GUY LEMEUNIER, F. C. (1979): Materiales para una historia del Reino de Murcia en los tiempos modernos, EDITUM.
- McCAA, R. (1984): "Calidad, clase, and marriage in colonial Mexico: the case of Parral, 1788-90," *The Hispanic American Historical Review*, 3, 477–501.
- MOLINA-PUCHE, S. (2005): "Las élites locales del corregimiento Chinchilla-Villena en el siglo XVII," Ph.D. thesis, Departamento de Historia Moderna, Contemporánea y de América de la Universidad de Murcia.
- Obispado de Cartagena (1909): "Boletín Eclesiástico del obispado de Cartagena de 1908," Butsem y Compañía.
- OLIVETTI, C. AND M. D. PASERMAN (2015): "In the Name of the Son (and the Daughter): Intergenerational Mobility in the United States, 1850–1940," *American Economic Review*, 105, 2695–2724.
- OLIVETTI, C., M. D. PASERMAN, AND L. SALISBURY (2018): "Three-generation mobility in the United States, 1850-1940: The role of maternal and paternal grandparents," *Explorations in Economic History*, 70, 73–90.
- PARMAN, J. (2011): "American mobility and the expansion of public education," *The Journal* of Economic History, 71, 105–132.
- PÉREZ LEÓN, J. (2012): "Hidalgos indianos ante la Real Chancillería de Valladolid: el caso peruano en época de los Borbones," Ph.D. thesis, Universidad de Valladolid.

- PÉREZ ORTÍZ, G., F. GONZÁLEZ LOZANO, AND A. VIVAS MORENO (2017): "La investigación genealógica en los archivos eclesiásticos: fuentes y metodología," *Journal of information and documentation systems*, 11, 41–50.
- PITA PICO, R. (2013): "Nobleza y limpieza de sangre en el nororiente neogranadino, 1710-1810: entre la segregación y la amenaza étnica," *Procesos históricos*, 24, 16–43.
- SANTIAGO-CABALLERO, C. (2018): "Social mobility in nineteenth century Spain, Valencia, 1841-1870," Working papers in Economic History, Universidad Carlos III de Madrid.
- SOLON, G. (2018): "What do we know so far about Multigenerational Mobiliy?" *Economic Journal*, 128, 340–352.
- SORIA MESA, E. (2004): "Genealogía y poder. Invención de la memoria y ascenso social en la España Moderna," *Revista de Historia Moderna*, 30, 21–56.
- Тома́s Faci, G. (2015): "Geografía de la población infanzona en Aragón (ss. XIII-XV)," Aragón en la Edad Media, 26, 321–349.

Parish	Num Marriages	Frac. High Status		Match Rate	
		Grooms	Brides	Grooms	Brides
San Antolin	2927	8.06	9.90	14.7	19.4
San Juan	1332	7.50	8.93	11.1	13.6
San Lorenzo	562	7.82	9.60	5.33	6.04
San Miguel	1447	8.91	10.2	15.6	19.2
San Pedro	966	11.8	16.4	9.73	15.2
Santa Catalina	774	20.2	23.3	10.3	13.8
Santa Maria	10167	3.72	4.58	20.2	25.7
TOTAL	18175	6.37	7.79	16.8	21.6

Table 1: Summary Statistics by Parish

Notes: Parish is the church in Murcia in which the marriages were performed. Santa Maria is the main cathedral church. Fraction high status denotes the percentages of grooms and brides who were denoted as having status as a *don* or *doña*. Match rate is the percent of marriages for which the marriage of the parents of the groom or bride repsectively could be linked to the child's marriage.

		0	
	(1)	(2)	(3)
Panel A: Grooms (N=18175)			
Father	0.790***		0.586***
	(0.009)		(0.019)
Mother		0.710***	0.278***
		(0.010)	(0.019)
Parish FE	Yes	Yes	Yes
R-squared	0.636	0.516	0.670
	(1)	(2)	(3)
Panel B: Brides (N=18175)			
Father	0.719***		0.317***
	(0.009)		(0.019)
Mother		0.791***	0.561***
		(0.011)	(0.023)
Parish FE	Yes	Yes	Yes
R-squared	0.537	0.636	0.684

Table 2: Status Transmission: Marriage Records

Notes: OLS regression for status as a *don* or *doña*. Dependent variable is the standardized social status measure, measured at the time of marriage. The independent variables are the standardized social status measures for fathers and mothers respectively, measured at the time of their child's marriage. All standard errors are clustered at the parish level. *, **, and *** represent p < 0.1, p < 0.05, and p < 0.01 respectively.

				0			
	Ν	Middle Ag	e	Young			
	(1)	(2)	(3)	(4)	(5)	(6)	
Panel A: Grooms (N=3070)							
Father	0.624***		0.419**	0.687***		0.585***	
	(0.061)		(0.118)	(0.029)		(0.053)	
Mother		0.589***	0.321**		0.559***	0.144**	
		(0.067)	(0.124)		(0.054)	(0.044)	
Parish FE	Yes	Yes	Yes	Yes	Yes	Yes	
R-squared	0.415	0.374	0.474	0.495	0.340	0.504	
	(1)	(2)	(3)	(4)	(5)	(6)	
Panel B: Brides (N=3931)							
Father	0.427***		0.219***	0.497***		0.362***	
	(0.040)		(0.037)	(0.043)		(0.061)	
Mother		0.494***	0.369***		0.444***	0.218***	
		(0.057)	(0.063)		(0.041)	(0.034)	
Parish FE	Yes	Yes	Yes	Yes	Yes	Yes	
R-squared	0.223	0.280	0.312	0.279	0.228	0.307	

Table 3: Status Transmission in Linked Marriage Records

Notes: OLS regression for status as a *don* or *doña*. Dependent variable is the standardized social status measure. The independent variables are the standardized social status measures for fathers and mothers respectively. "Middle Age" represents status measured for both children and parents at when both generations have a child getting married. "Young" represents status measured for both children and parents at the time of each's own marriage respectively. All standard errors are clustered at the parish level. *, **, and *** represent p < 0.1, p < 0.05, and p < 0.01 respectively.

			1	
	(1)	(2)	(3)	(4)
Panel A: Grooms (N=3070)				
Father	0.625***	0.777***	0.560***	0.551***
	(0.053)	(0.066)	(0.060)	(0.059)
Mother	0.280***		0.271***	0.277***
	(0.043)		(0.046)	(0.045)
Paternal GF		0.127**	0.115**	0.105^{*}
		(0.049)	(0.046)	(0.052)
Paternal GM				-0.023
				(0.020)
Maternal GF				0.065
				(0.053)
Maternal GM				-0.037
				(0.029)
Parish FE	Yes	Yes	Yes	Yes
N	3070	3070	3070	3070
R-Squared	0.769	0.755	0.777	0.779
	(1)	(2)	(3)	(4)
Panel B: Brides (N=3931)				
Father	0.258***	0.721***	0.243***	0.239***
	(0.038)	(0.026)	(0.039)	(0.037)
Mother	0.615***		0.613***	0.632***
	(0.051)		(0.050)	(0.037)
Paternal GF		0.043	0.032*	0.034
		(0.024)	(0.016)	(0.022)
Paternal GM				0.002
				(0.048)
Maternal GF				0.026
				(0.028)
Maternal GM				-0.059**
				(0.020)
Parish FE	Yes	Yes	Yes	(0.020) Yes
Parish FE N	Yes 3931	Yes 3931	Yes 3931	(0.020) Yes 3931

Table 4: Status Transmission from Grandparents

Notes: OLS regression for status as a *don* or *doña*. Dependent variable is the standardized social status measure, measured at the time of marriage. The independent variables are the social status measures for the relatives shown, measured in "middle age" as defined in Table 3. All standard errors are clustered at the parish level. *, **, and *** represent p < 0.1, p < 0.05, and p < 0.01 respectively.

Table 5: Marriage Order Regressions							
	(1)	(2)	(3)	(4)	(5)	(6)	
Panel A: Grooms (N=3070)							
Father	0.856***	0.625***	0.624***	0.625***	0.625***	0.625***	
	(0.039)	(0.053)	(0.053)	(0.053)	(0.054)	(0.054)	
Mother		0.280***	0.280***	0.280***	0.280***	0.280***	
		(0.043)	(0.043)	(0.043)	(0.044)	(0.044)	
Number of siblings		-0.001		0.002		0.002	
		(0.003)		(0.008)		(0.008)	
Number own sex siblings			-0.004	-0.007		-0.009	
			(0.004)	(0.013)		(0.015)	
First married own sex					0.009	0.013	
					(0.019)	(0.023)	
Parish FE	Yes	Yes	Yes	Yes	Yes	Yes	
R-Squared	0.746	0.769	0.769	0.769	0.769	0.769	
	(1)	(2)	(3)	(4)	(5)	(6)	
Panel B: Brides (N=3931)							
Father	0.742***	0.258***	0.258***	0.258***	0.258***	0.258***	
	(0.016)	(0.038)	(0.038)	(0.038)	(0.038)	(0.038)	
Mother		0.614***	0.615***	0.614***	0.614***	0.614***	
		(0.051)	(0.051)	(0.051)	(0.051)	(0.051)	
Number of siblings		-0.006		-0.009		-0.009	
		(0.004)		(0.005)		(0.005)	
Number own sex siblings			-0.005	0.005		0.002	
_			(0.005)	(0.006)		(0.005)	
First married own sex					0.016*	0.022***	
					(0.007)	(0.006)	
Parish FE	Yes	Yes	Yes	Yes	Yes	Yes	
R-Squared	0.570	0.704	0.704	0.704	0.704	0.704	

Notes: OLS regression for status as a don or doña. Dependent variable is the standardized social status mea-
sure, measured at the time of marriage. The independent variables are the standardized social status mea-
sures for fathers and mothers respectively, measured at the time of their child's marriage. Marriage order
based on sample of marriages matched back one generation. All standard errors are clustered at the parish
level. *, **, and *** represent $p < 0.1$, $p < 0.05$, and $p < 0.01$ respectively.
represented to the present p < 0.00 p < 0.00 respectively.

Table 5: Marri	age Order F	Perressions

Table 0. Tumber of Observed Siblings based on Farents Status							
	Groo	oms	Brid	des			
	High Status	Low Status	High Status	Low Status			
Number of Brothers	0.28	0.68	0.38	0.64			
Number of Sisters	0.37	0.82	0.64	0.87			
Number of Siblings	0.65	1.50	1.02	1.51			

Table 6: Number of Observed Siblings Based on Parents' Status

Notes: Table displays the number of other marriages from children of the same parents based on the status of parents, measured at the time of the parents marriage.

Table 7: Assortative Mating							
	(1)	(2)	(3)	(4)			
Panel A: Grooms (N=18175)							
Bride Status	0.796***	0.578***	0.747***	0.616***			
	(0.016)	(0.015)	(0.022)	(0.019)			
Bride's Father's Status		0.300***		0.321***			
		(0.015)		(0.014)			
Bride's Mother's Status			0.062*	-0.067*			
			(0.026)	(0.028)			
Parish FE	Yes	Yes	Yes	Yes			
R-squared	0.640	0.682	0.641	0.683			
	(1)	(2)	(3)	(4)			
Panel B: Brides (N=18175)							
Groom Status	0.792***	0.702***	0.610***	0.613***			
	(0.005)	(0.015)	(0.011)	(0.013)			
Groom's Father's Status		0.114***		-0.006			
		(0.015)		(0.023)			
Groom's Mother's Status			0.256***	0.259***			
			(0.011)	(0.017)			
Parish FE	Yes	Yes	Yes	Yes			
R-squared	0.641	0.646	0.673	0.673			

Notes: OLS regression for status as a *don* or *doña*. Dependent variable is the standardized social status measure, measured at the time of marriage. The independent variables are the social status measures for the relatives shown, at the time of the child's marriage. All standard errors are robust. *, **, and *** represent p < 0.1, p < 0.05, and p < 0.01 respectively.





Notes: Own elaboration with information using the *Boletín Eclesiástico* from the Cargatena Bishopric from 1908, pages 89-94 (Obispado de Cartagena, 1909). Because some streets have changed names since 1908, we used historical maps of Murcia from 1896 and 1882 (Archivo General Región de Murcia, 2019).

Figure 2: Example of marriage record. Parish of Santa María (Murcia) in 1761.

tanched comba

Ignacio Soto con Doña Francisca Sánchez, Desposados

En la ciudad de Murcia en veinte y ocho días de mes de Diciembre de mil setecientos sesenta y uno y presente Don Antonio García, cura teniente de esta Iglesia parroquial de Santa María, despose por palabras de presente, que hacen verdadero matrimonio a Ignacio Soto, natural y feligrés de San Bartolomé, hijo de Santiago Soto y de Doña Josefa del Castillo, con Doña Francisca Sánchez, natural y feligresa de esta Parroquia, hija de Don Ginés Sánchez y Doña Mariana Mateos. [...] En fecha de dicho día por ante Francisco Bueno, Notario mayor fueron testigos Don Antonio Amador, Presbítero, Don Ignacio de Reyabas, Don Antonio Casanoba y en fe de ello lo firme - Don Antonio García

Ignacio Soto with Doña Francisca Sánchez, Married

In the city of Murcia December 28, 1761, under the presence of *Don* Antonio García, the local priest of this parish church of Santa María, marry out of their own will, in true matrimony, Ignacio Soto, born and parishioner of San Bartolomé, son of Santiago de Soto and *Doña* Josefa del Castillo, with *Doña* Francisca Sánchez, born and parisioner of this Parish, daughter of *Don* Ginés Sánchez and *Doña* Mariana Mateos. [...] At the said date, in the presence of Francisco Bueno, Notary Major, as witnesses *Don* Antonio Amador, presbyter, *Don* Ignacio de Reyabas, *Don* Antonio Casanoba and giving faith about all signed - *Don* Antonio García



Notes: Source: *España, registros parroquiales y diocesanos, 1307-1985*. Database with images, FamilySearch. Transcription from the original. In the transcription we have substituted the abbreviations in the original source with the complete words. Our translation to English from the original. Lower figure shows family relationship.



Figure 3: Coefficients on β_F and β_M by Decade.

Notes: Coefficients from OLS regression for status as a *don* or *doña* for brides and grooms, separately by decade. Dependent variable is the standardized social status measure, measured at the time of marriage. The independent variables are the standardized social status measures for fathers and mothers respectively, measured at the time of their child's marriage. Plots show 95% confidence intervals based on robust standard errors and including parish fixed effects. Date refers to the first year of the decade; i.e., 1700 is for 1700-1709. "Short" regression refers to a regression of the status of the groom or bride on the father only, excluding the mother. Assorative mating refers to the coefficient on a regression of the status of the parents for the indicated relative, including fixed effects.

A Appendix

A.1 Hidalgos and dons

In Spain, the term *hidalgo* refers to the low aristocracy. It is typically reserved for people who are of noble origin but do not hold any nobility title. Dewald (1996) notes that during the 18th century there were two regions in Europe with different patterns regarding the number of aristocrats: Central Europe, with a percentage lower than 1%, and the periphery (Spain, Poland, Hungary and England), with more than 5%. Madramany y Calatayud (1788) mentioned that *hidalgos* were originally from "good places," meaning places with little Arab influence in the north of Spain (Galicia, Asturias, Cantabria and the Basque Country). Therefore, lineage and place of origin, explain the high percentage of *hidalgos* in northern Spain. Moreover, the percentage in the remaining of Spain are closer to the 1% in central Europe. It is the high percentage in northern Spain makes the overall average closer to peripheral Europe.

According to the Floridablanca census (1787), 4.6% of the population in Spain were *hidalgos*. The northern provinces have over 10%, whereas Extremadura and Andalusia in the South have less than 1%. In the Kingdom of Aragon, there was traditionally a lower presence of *hidalgos* (Tomás Faci, 2015): 1.5% in Aragon, 0.14% in Valencia and 0.16% in Catalonia (INE, 2019). In the Region of Murcia, which belongs to the Kingdom of Castile, but shares many characteristics with Valencia and Catalonia, the number is 1.80%. In the city of Murcia, which is the interest of the article, the percentage of *hidalgos* was 0.45%. The fraction of *dons* that we observed in the data is an order of magnitude higher. This means, consistent with our interpretation, that the term *don*, by the 18th century, was no longer reserved for the nobility. On the contrary, the *nouveau riche*, the clergy and the higher ranks of the military also hold the honorary title. Dewald (1996) notices the sharp reduction in the percentage of aristocrats in Europe, beginning in the 18th century. In Spain, we observe a sharp decline between 1768 and 1787, from 7.9% to 4.6% (INE, 2019). In our

data, however, we do not observe a rise in the fraction of brides or grooms carrying the honorific title of *don or doña*.

A.2 Data description

Table A1 shows the fraction of the grooms and brides with the honorific, separately based on the status of the parents of the marrying couples. Approximately 6% of grooms and 8% of brides are recorded as high status. It is immediately clear that there is a high degree of persistence in this measure of social status. Of the 890 grooms with both a high status father and mother, 89% of them were themselves a *don* at marriage. This fraction is similar for brides. Conversely, only about 1% of the much larger number of individuals with low status parents were themselves *dons* or *doñas*. In general, the transmission from parents to grooms versus that to brides is similar for various combinations of status of parents, with one major exception. If the father is low status but the mother is high status, then brides are much more likely to be high status than are grooms.

In Table A2, we show the fraction of grooms and brides which marry a high status partner, based on their own status and that of their own parents. Of grooms listed with *don* status, 90% are marrying brides who are *doñas*. The comparable number for *doñas* is 73%. The parents' status matters in mating above their effect on the status of their children. In the first row, for example, we can see that while 92% of *don* grooms whose parents were themselves high status marry a high status *doña* woman, only 75% of *dons* whose parents were both low status have a high status bride. For women this difference is even stronger: While 96% of *doña* brides with high status parents marry a *don*, only 35% of "upwardly mobile" *doñas* with low status parents marry someone of high status. This is true for low status individuals as well: A downwardly mobile groom with high status grooms with low status preveals several interesting facts about assortative mating. Most obviously, much like the high intergenerational transmission of status, assortative

mating on status is very strong. Second, and much less obviously, the degree of assortative mating depends on the status of the parents as well as that of the children. That is, a high status groom and particularly a high status bride who comes from higher status parents is more likely to marry a high status individual than a high status individual with low status parents. Goñi (2015) shows how assortative mating among the high classes of England increased inequality, but we are not aware of any study of mating among the low classes.

A.3 Linking

In this section, we describe the procedure we used to link marriage records from one generation to the previous generation. The only link variables are names: The names of both parents of either the bride or groom (in the childrens' generation), and the names of the marrying couple (in the parents' generation). For each generation, we restrict the possible set of links in the parents generation to the set of marriages between 18 and 45 years before the marriage of the children; in other words, we assume that individuals marry between ages 18 and 45.

The linking is performed iteratively, similar to Ferrie (1996) and Abramitzky et al. (2014). In each step, the links are removed from the set of both parents and children linked later. In all links, we first perform a minimal set of cleaning after parsing the names into given names and surnames.¹¹ This is limited to stripping out accents and the \tilde{n} and removing the words "Las", "Los", "La", "De", "Del", and "De La".

In the first step, we link on exact names, *i.e.*, we require the spelling to be identical, on four words: the given name and the surname of the parents, and disregard multiple links. Multiple links are relatively rare given the small area from which the marriages come and the use of four names to link. In the second step, we link replacing the names as transcribed and cleaned with a set of standardized spellings, and then again link on all four

¹¹This parsing is non-trivial, with the number of names for individuals varying and some name, such as Garcia, potentially being either a given name or a surname. We also exclude from the set of names special constructions such as "De La Cruz" or "Del Carmen", which are not consistently recorded for individuals.

words. For example, for surnames both "Hespinosa" and "Hespinossa" are standardized to "Espinosa", and for given names both "Josef" and "Joseph" are standardized to "Jose." Finally, we use a set of standardized phonetic changes to clean the names, and repeat linking all four names. These substitutions were as follows: "Y" for "I" or "LL", "B" for "V", "C" for "Z", "F" for "Ph", "T" for "Th" or "TT", "G" for "J" or "X", "S" for "CH" or "SS", "N" for "NN" or "Ñ", "HU" for "GÜ", "Q" for "Qu", and removing any remaining "H".¹² It is worth noticing that the same individual could sometimes appear as *Fernandez* and other times as *Hernandez*. In those, more intricate cases, we have resorted to linking by hand before our iterative process.

After the second step, we used a novel third step. We look for the small number of remaining multiple links (number). In this final population of multiple links, there is still some remaining information. In particular, we have many marriages where all the potential links all have the same status information on all relevant variables. For the regressions using grandparents, that means that the status as a *don* is the same for all six individuals listed in the candidate link. For the groom's regressions these six individuals are: the the groom's parents and grandparents. Notice, that we are not requiring those to be the same to the original marriage. As we explain above, status could change over time. Given the nature of the data, there are $2^6 = 64$ possible combinations, *i.e.*, we have 6 individuals that could have a binary value.¹³ In practice, there is high correlation among the values for those 6 individuals and some combinations are extremely rare. Moreover, because *don* status is held by a relatively small minority of the population, the group where all 6 individuals are not *don* makes up a substantial fraction of the relatively small number of multiple links. This is particularly important in this, and other settings, where usually the individuals with low status are also those with common names, and they are out of the

¹²In our experience, using *Soundex* with Spanish names did not perform as well. This is not surprising, because *Soundex* is built for Germanic languages, like English and German, which are phonetically based on consonants. Latin languages, like Spanish and Italian, are phonetically based on vowels.

¹³For the regressions using only the status of the parents, but taken at two different points in the lifecycle, we would only need that the status of both the father and the mother of the groom to be the same in all candidates. In this case there are only 4 combinations.

linked sample. Therefore, adding these final links helps in reducing the selection incurred when linking on names.

The great majority of the links are exact. Of the combined 7,017 links for grooms and brides, 6,832 (97.4%) are exact. There were a further 41 (0.6%) links after standardizing the spelling of names, and 48 links (0.7%) after making phonetic substitutions. Additionally, there were a further 96 (1.4%) multiple links for which the status of all six individuals was identical across multiple links and so could be included in the sample. As mentioned above, even if only 1.4% of the final links were included using the last step, those are all links referring to people with common names.

In the original dataset, we have 18,175 grooms and brides. In the first stage, when we link directly on exact names, we end up with 6,832. After the second stage, when we link the remaining observations using phonetically clean names, we end up with 6,832+41+48=6,921 matches, and 139 observations with multiple candidate matches. Of those 139 matches we are able to create 96 synthetic observations, because all the candidate matches have the same values for status on all individuals. Of those 96 synthetic observations, in 89 of the cases none of the 6 individuals in all the multiple candidates have the don status, and in 7 cases all 6 individuals have the don status.

A.4 Surnames

In this subsection, we study social mobility using the information contained in last names using an alternative method to studying social mobility prominently associated with Clark (2014). This technique looks at the relative representation of surnames within the elite, over time. That is, rather than rely on linked data, one regresses the share of individuals who possess some measure of high (or low) status among people with a particular surname on the share of individuals with that surname having high (or low) status in a previous generation. This avoids the need to link data. Clark (2014) argues that it also generates a better measure of an underlying measure of social status with high intergenerational persistence

than the use of more standard measures such as income and occupation, which can be measured with considerable individual error. He generally finds estimates of between 0.7 and 0.8 across different societies, a pattern he names the "iron law of social mobility".

To compare our estimates with his, we take the linked sample so that we can measure status for parents and children at consistent ages, to make the comparison as similar as possible. We take the "young" measure of status; that is, we use the measures of status taken at a groom or bride's marriage and that of their parents, as in columns (4) through (6) of Table 3. For each observation, we calculate the *share* of individuals with a particular surname with status as a *don* or *doña* in each generation, and regress one on the other. The regressions are weighted by the number of individuals with each surname, to maximize comparison with the regression from the linked data.

The results from this regression are displayed in Table A3. The numbers for fathers are relatively similar to those from individual regressions, although they are still not in excess of 0.7. However, surname based measures seem to give considerably lower estimates for the effect of mothers on social status on grooms and brides. This is not an artifact of naming conventions: For both grooms and brides, someone with the surname "Navarro" will have a father with this surname, and a mother with a different surname. Remember that in Spain, women do not change their name when they marry. The low values in the coefficients in Table A3, especially for women, contrast with the results in Solon (2018) that pseudo-links would tend to inflate the estimated intergenerational elasticity because they measure group, rather than individual, transmission of status.

Surname-based methods appear in this case to give little information as to the role of women in social mobility, even when they are feasible to use. Comparing columns (1) and (2), and similarly for columns (4) and (5), we can see the drop in the size of the coefficient, and an R-squared that is an order of magnitude smaller. In other words, columns (2) and (5) would seem to imply that mothers status have very little effect on their children's status. This result underscores the importance of having not only information on women's status,

but that this information is about the individual women. If we were to have the surname for all the women in our sample, but not their individual status, based on the results of Table A3, we would have wrongly concluded that they played little role in determining the status of their sons and daughters.

Table A1. Tercent Dons by Talents Status								
Relationship between Father and Mother								
All Don/Doña X/X Don/X X/Doña								
Groom	6.37	89.6	.955	71.2	30.0			
Bride	7.79	92.7	1.26	67.1	68.3			
Number Grooms	18175	859	16745	139	432			
Number Brides	18175	852	16712	140	471			

Table A1: Percent Dons by Parents Status

Notes: Value in the table is the percent of grooms and brides with Don/Doña status based on the status of their (own) parents, given in the column header as the status of the father and status of the mother. Number grooms/brides refers to the total number of grooms and brides respectively with the parental status combination shown above.

	Relationship between Father and Mother						
	All	Don/Doña	X/X	Don/X	X/Doña		
		Percent Sp	ouse Hi	gh Status			
Groom Don	89.9	92.3	75.6	90.9	92.3		
Groom X	2.20	30.3	1.36	7.50	39.0		
Bride Doña	73.5	96.3	34.7	96.7	50.4		
Bride X	.698	48.7	.237	19.5	5.15		
		NT 1	6.0				
		Numb	er of Sp	ouses			
Number Groom Don	1159	770	160	99	130		
Number Groom X	17016	89	16585	40	302		
Number Bride Doña	1417	738	348	93	238		
Number Bride X	16758	121	16397	46	194		

Table A2: Percent Spouse Don/Doña Based on Parent's and Own Statuses

Notes: Value in the table is the percent of spouses with Don/Doña status based on the combination of the status of their (own) parents, given in the column header as the status of the father and status of the mother, and their own status. Number grooms/brides refers to the total number of grooms and brides respectively with the parental status combination shown above.

Table A3: Surnames Regression							
	(1)	(2)	(3)	(4)	(5)	(6)	
Rel. share groom father	0.655***		0.651***				
	(0.044)		(0.043)				
Rel. share groom mother		0.242***	0.022				
		(0.032)	(0.021)				
Rel. share bride father				0.368***		0.349***	
				(0.012)		(0.016)	
Rel. share bride mother					0.223***	0.077**	
					(0.041)	(0.028)	
Parish FE	Yes	Yes	Yes	Yes	Yes	Yes	
Ν	3070	3070	3070	3931	3931	3931	
R-Squared	0.615	0.068	0.616	0.325	0.087	0.332	

D. T11 AO C .:

Notes: OLS regression for status as a *don* or *doña* based on surname status. Dependent variable is the fraction of individuals with a surname with high status, and the independent variables are these fractions for the surnames of an individuals father and mother. Status is measured at the time of marriage for both generations. Regressions are weighted by the number of individuals in the younger generation with a particular surname. All standard errors are clustered at the parish level. *, **, and *** represent p < 0.1, p < 0.05, and p < 0.01 respectively.